Author's response to reviews

Title: Ethical challenges related to elder care. High level decision-makers experiences

Authors:

Anna-Greta Mamhidir (anna-greta.mamhidir@lg.se)
Mona Kihlgren (mona.kihlgren@orebroll.se)
Venke Sorlie (venke.sorlie@medisin.uio.no)

Version: 2 Date: 25 August 2006

Dear Emma Parkin,
Associate Editor, BMC Medical Ethics

According to your suggestion, I have made a revision and combined the originally two Manuscripts into one and hope it will be accepted for publication in BMC Medical Ethics. This new Manuscript is entitled "Ethical challenges related to elder care. High level decision-makers' experiences"

The former submitted Manuscripts were entitled - "Ethical dilemmas in elder care as narrated by high-level decision-makers (MS: 1330616190109001) and Being in ethically difficult situations related to elder care: High level decision-makers' experiences" (MS: 1370148345109003).

If this new Manuscript is of interest, I would appreciate a prompt replay since my doctoral dissertation is scheduled within the year.

Yours sincerely
Anna-Greta Mamhidir, PhD-student
Correspondence address: Anna-Greta Mamhidir, Nylandsgatan 16 SE-826 31 Soderhamn, Sweden
E-mail: anna-greta.mamhidir@lg.se
Telephone: +46 270 158 47, +46 70 664 28 17